Author's response to reviews

Title: High-resolution SNP array analysis of patients with developmental disorder and normal array CGH results

Authors:

Linda Siggberg (linda.siggberg@helsinki.fi)
Sirpa Ala-Mello (sirpa.alamello@fimnet.fi)
Tarja Linnankivi (tarja.linnankivi@hus.fi)
Kristiina Avela (kristiina.avela@vaestoliitto.fi)
Ilari Scheinin (ilari.scheinin@helsinki.fi)
Kati Kristiansson (kati.kristiansson@thl.fi)
Päivi Lahermo (päivi.lahermo@helsinki.fi)
Marja Hietala (marja.hietala@tyks.fi)
Liisa Metsahonkala (liisa.metsahonkala@hus.fi)
Esa Kuusinen (esa.kuusinen@satshp.fi)
Maarit Laaksonen (maarit.laaksonen@thl.fi)
Janna Saarela (janna.saarela@helsinki.fi)
Sakari Knuutila (sakari.knuutila@helsinki.fi)

Version: 6 Date: 18 August 2012

Author's response to reviews:

Dear editors,

I received an email asking me to edit the style of written English in the manuscript. The manuscript has been reviewed by a native-English language professional, a service provided by the University of Helsinki, already before the last submission, as this was one of the requirements of the reviewers. I tried to enquire whether I was to do it again by sending an email to Mr Declaro, unfortunately I did not receive an answer.

If you feel that the language needs to be further edited I will see to it promptly.

kind regards,
Linda Siggberg